Author’s response to reviews

Title: Bilateral pleural effusion and interstitial lung disease as unusual manifestations of Kikuchi-Fujimoto disease: case report and literature review

Authors:

ALBERTO GARCIA-ZAMALLOA (ALBERTO.GARCIA-ZAMALLOAZAMALLOA@OSAKIDETZA.NET)
JORGE TABOADA (JORGE.TABOADAGOMEZ@OSAKIDETZA.NET)
PILAR BERNARDO (PILAR.BERNARDOGALAN@OSAKIDETZA.NET)
MAGDALENA FERNANDEZ (MAGDALENA.FERNANDEZMARTINEZ@OSAKIDETZA.NET)
LAURA ZALDUMBIDE (LAURA.ZALDUMBIDEDUENAS@OSAKIDETZA.NET)
MARIO UGARTE (MARIO.UGARTEMAIZTEGUI@OSAKIDETZA.NET)

Version: 5 Date: 21 September 2010

Author’s response to reviews:

Dear Sir:

My colleagues and myself have just reviewed by second time the manuscript, following the instructions and recommendations of the doctor Kyung-Yil Lee. In our opinion, we have obtained some new improvement.

We decided to include a fourth figure with histopathological findings and also all the figure legends (previously we did not find the way to perform it). We also corrected the cite numbers of references and their format.

Following your recommendation, we sent the manuscript to a professional copyediting service (Hedapen Global Services). We hope you will find improved the language quality of our manuscript.

Best regards

Alberto Garcia-Zamalloa